Functional relationships reveal differences in the water cycle representation of global water models

Sebastian J. Gnann^{a,*}, Robert Reinecke^a, Lina Stein^a, Yoshihide Wada^{b,c}, Wim Thiery^d, Hannes Müller Schmied^{e,f}, Yusuke Satoh^g, Yadu Pokhrel^h, Sebastian Ostbergⁱ, Aristeidis Koutroulis^j, Naota Hanasaki^k, Manolis Grillakis^j, Simon N. Gosling^I, Peter Burek^c, Marc F. P. Bierkens^{m,n}, and Thorsten Wagener^a

^aInstitute of Environmental Science and Geography, University of Potsdam, Potsdam, Germany

^bClimate and Livability, Biological and Environmental Science and Engineering Division, King Abdullah University of Science and Technology, Thuwal, Saudi Arabia

^cInternational Institute for Applied Systems Analysis, Laxenburg, Austria

^dDepartment of Hydrology and Hydraulic Engineering, Vrije Universiteit Brussel, Brussels, Belgium

^eInstitute of Physical Geography, Goethe University Frankfurt, Frankfurt am Main, Germany

^fSenckenberg Leibniz Biodiversity and Climate Research Centre (SBiK-F), Frankfurt am Main, Germany

^gMoon Soul Graduate School of Future Strategy, Korea Advanced Institute of Science and Technology, Korea

^hDepartment of Civil and Environmental Engineering, Michigan State University, East Lansing, MI, USA

ⁱPotsdam Institute for Climate Impact Research (PIK), Member of the Leibniz Association, Potsdam, Germany

^jSchool of Chemical and Environmental Engineering, Technical University of Crete, Greece

^kNational Institute for Environmental Studies, Tsukuba, Japan

School of Geography, University of Nottingham, Nottingham, United Kingdom

^mDepartment of Physical Geography, Utrecht University, The Netherlands

ⁿUnit Soil and Groundwater Systems, Deltares, Utrecht, The Netherlands

*Correspondence: gnann1@uni-potsdam.de

Functional relationships reveal differences in the water cycle representation of global water models

Sebastian J. Gnann^{a,1,2}, Robert Reinecke^{a,1}, Lina Stein^a, Yoshihide Wada^{b,c}, Wim Thiery^d, Hannes Müller Schmied^{e,f}, Yusuke Satoh^g, Yadu Pokhrel^h, Sebastian Ostbergⁱ, Aristeidis Koutroulis^j, Naota Hanasaki^k, Manolis Grillakis^j, Simon N. Gosling¹, Peter Burek^c, Marc F. P. Bierkens^{m,n}, and Thorsten Wagener^a

^a Institute of Environmental Science and Geography, University of Potsdam, Potsdam, Germany; ^bClimate and Livability, Biological and Environmental Science and Engineering Division, King Abdullah University of Science and Technology, Thuwal, Saudi Arabia; ^c International Institute for Applied Systems Analysis, Laxenburg, Austria; ^d Department of Hydrology and Hydraulic Engineering, Vrije Universiteit Brussel, Brussels, Belgium; ^eInstitute of Physical Geography, Goethe University Frankfurt, Frankfurt am Main, Germany; ^f Senckenberg Leibniz Biodiversity and Climate Research Centre (SBiK-F), Frankfurt am Main, Germany; ^gMoon Soul Graduate School of Future Strategy, Korea Advanced Institute of Science and Technology, Korea; ^hDepartment of Civil and Environmental Engineering, Michigan State University, East Lansing, MI, USA; ⁱ Potsdam Institute for Climate Impact Research (PIK), Member of the Leibniz Association, Potsdam, Germany; ^jSchool of Chemical and Environmental Engineering, Technical University of Crete, Greece; ^kNational Institute for Environmental Studies, Tsukuba, Japan; ¹School of Geography, University of Nottingham, United Kingdom; ^mDepartment of Physical Geography, Utrecht University, The Netherlands; ⁿUnit Soil and Groundwater Systems, Deltares, Utrecht, The Netherlands

This manuscript was compiled on December 2, 2022

Global water models are widely used for policy-making and in scientific studies, but substantial inter-model differences highlight the 2 need for additional evaluation. Here we evaluate global water models by assessing so-called functional relationships between system forcing and response variables. The more widely used comparisons between observed and simulated fluxes provide insight into model 6 behavior for the representative area of an observation, and can therefore potentially improve the model for that area. Functional relation-8 ships, by contrast, aim to capture how system forcing and response 9 variables co-vary across large scales, and thus offer the potential for 10 model improvement over large areas. Using 30-year annual averages 11 from 8 global water models, we quantify such functional relation-12 ships by calculating correlations between key forcing variables (pre-13 cipitation, net radiation) and water fluxes (actual evapotranspiration, 14 groundwater recharge, total runoff). We find strong disagreement 15 for groundwater recharge, some disagreement for total runoff, and 16 the best agreement for evapotranspiration. Observation- and theory-17 derived functional relationships show varying agreements with mod-18 els, indicating where model representations and our process un-19 derstanding are particularly uncertain. Overall, our results suggest 20 that model improvement is most important for the representation of 21 energy balance processes, recharge processes, and generally for 22 model behavior in dry and cold regions. We argue that advancing 23 our ability to simulate global hydrology requires a better perceptual 24 understanding of the global water cycle. To evaluate if our models 25 match that understanding, we should explore alternative evaluation 26 strategies, such as the use of functional relationships. 27

global hydrological models | land surface models | model evaluation | rank correlations | global hydrology

G lobal water models – including hydrological, land surface, and dynamic vegetation models (1) – inform water management policies. Many global modeling studies explicitly aim to provide policy-relevant information (e.g. 2–6). The Sixth Assessment Report (AR6) of the Intergovernmental Panel on Climate Change (IPCC) (7) draws heavily on results from global water models, which provide information on the impacts of climate change on streamflow (8, 9), terrestrial water storage (10), and groundwater recharge (11). Some of these models are embedded in global water information services to provide water information to a wide array of stakeholders. For instance, the Global Groundwater Information System (12) shares information required for sustainable groundwater resources development and management. The Aqueduct framework (13) calculates risk indicators to derive water risk maps valuable for companies, governments, and non-governmental organizations. And the African Flood and Drought Monitor (14) continuously predicts drought and flood indicators using various forecasting products.

11

12

13

14

15

16

17

18

19

Global water models have also become an essential tool in 20 Earth system science. Measurements of some hydrological vari-21 ables are very sparse and insufficient for large-scale analyses. 22 Hence we regularly use global water models to provide globally 23 coherent estimates of variables such as groundwater recharge 24 and groundwater storage change (15-17). These model out-25 puts are often the basis for other studies, e.g. by providing 26 groundwater recharge as input to groundwater models (e.g. 27

Significance Statement

Global water models inform water management policies and are a cornerstone of Earth system science. Since global water models are increasingly used for projections of environmental change impacts, adequate methods to evaluate these models are imperative. Here we evaluate model behavior by comparing large-scale functional relationships between system drivers (climate forcing) and simulated and observed system outputs (water fluxes). We find substantial variability between models, and disagreements with observation-based functional relationships. For example, some models show a very strong relationship between groundwater recharge and precipitation, while others do not. Thus, projected changes in precipitation would result in different groundwater recharge estimates across models. Existing disagreements underscore the need for adequate evaluation strategies and for multi-model approaches to embrace uncertainty.

¹S.J.G. contributed equally to this work with R.R.

²To whom correspondence should be addressed. E-mail: gnann1@uni-potsdam.de

18, 19), making subsequent study results dependent on the
reliability of model outputs. Global models also provide a virtual laboratory which is used to assess the impacts of climate

and land use change on the water cycle and on hydrological

extremes such as floods and droughts (e.g. 8, 9, 20, 21).

Model disagreements highlight the need for model evaluation.

Past studies have revealed substantial disagreements between 34 global water models, showing that estimates of both the current 35 distribution and future trends of key water cycle components 36 remain uncertain (11, 22-25). While some of that uncertainty 37 stems from projected or observed climatic forcing, consider-38 able uncertainty stems from global water models themselves 39 40 (8, 11, 22, 23, 26, 27). For instance, Beck et al. (23) found distinct inter-model performance differences when comparing 41 simulated and observed streamflow for 10 global water models 42 driven by the same meteorological forcing. Comparing long-43 term trends in modeled terrestrial water storage to GRACE 44 satellites, Scanlon et al. (24) found that global models gen-45 erally underestimate trends and even show opposite trends 46 in some parts of the world. Different system conceptualiza-47 tions, such as including karst-related subsurface heterogeneity, 48 can lead to very different groundwater recharge estimates for 49 current and potential future climates (28). It is therefore not 50 surprising that the IPCC's AR6 (7) concludes from an analysis 51 of currently available global water model projections that "un-52 53 certainty in future water availability contributes to the policy challenges for adaptation, for example, for managing risks of 54 water scarcity". 55

To address inter-model differences as a source of uncertainty, 56 it is imperative that we evaluate how, where and why models 57 differ. Evaluating global models is, however, challenging due to 58 limitations in data availability (spatial and temporal bias, data 59 quality) and scale mismatches between observations and model 60 outputs (29). These challenges are not easy to overcome, but 61 they should motivate us to seek model evaluation strategies 62 that are suitable for global water models and better utilize 63 the information contained in the observations we have. 64

Towards functional model evaluation. Most evaluation strate-65 gies compare model outputs to historical observations over the 66 footprint for which the observation is representative. This can 67 be at the plot (e.g. flux towers), the catchment (e.g. gauging 68 stations), or grid cell (e.g. gridded remote sensing products) 69 70 scale. Such approaches are necessary but not sufficient to 71 robustly evaluate global models (29). First, these approaches compare simulated and observed values location by location 72 (or catchment by catchment), even though observations in 73 one location might contain information about geographically 74 different, but hydrologically similar, locations. Thus we might 75 miss the opportunity to improve the model for more than one 76 77 location at a time. Second, relevant information for model evaluation might not just lie in comparing the magnitudes 78 of simulated and observed values of a variable at a single 79 location, but rather in how a model simulates the spatial dis-80 tribution of a variable (i.e. its relative differences). And third, 81 82 a comparison with historical observations does not guarantee that a model reliably predicts system behavior under changing 83 conditions (30). We think that an alternative approach can at 84 least partially overcome these three shortcomings. 85

In this alternative strategy, we focus on the effective functional behavior of models (31). Effective functional behavior might be characterized by the relationship between system 88 forcing and response variables. For example, the concept of 89 equilibrium climate sensitivity, which quantifies the warming 90 response to doubling carbon dioxide concentrations, is often 91 used to describe how severe climate change might be (32). 92 In hydrology, the sensitivity (or elasticity) of streamflow to 93 changing climatic boundary conditions (33, 34) has been used 94 to better understand how changes in forcing translate into 95 changes in streamflow. This sensitivity can be used as an in-96 dicator of how quickly future water availability might change 97 under a changing climate (35). Forcing-response relationships 98 like these might be derived from (or constrained by) observa-99 tions and theory, but also by expert knowledge, thus enabling 100 us to bring our perception of how a system functions into the 101 model evaluation process (36). 102

Here we use so-called functional relationships and explore 103 their potential for global water model evaluation. We define 104 the term functional relationship as a relationship between two 105 (or more) variables, such as forcing-response relationships, or 106 relationships between system states and fluxes. Ideally, such 107 relationships should be based on process knowledge and first 108 principles, but empirical relationships may serve as a useful 109 starting point. Functional relationships have been used, for 110 example, to analyze land surface model functioning (31), to 111 evaluate catchment models (37), to derive constraints for model 112 regionalization (38), or to calibrate large-scale hydrological 113 models (39, 40). These examples are, however, scattered 114 among the literature and have not yet been formalized into 115 an evaluation framework. In the following, we outline how 116 an evaluation using functional relationships might look like, 117 show how it can help to shed new light on model behavior, 118 and discuss next steps required to fully benefit from functional 119 relationship based evaluation. 120

We evaluate 8 global water models (CLM4.5 (41), CWatM 121 (42), H08 (43), JULES-W1 (44), LPJmL (45), MATSIRO (46), 122 PCR-GLOBWB (47), and WaterGAP2 (48)) from phase 2b 123 of the Inter-Sectoral Impact Model Intercomparison Project 124 (ISIMIP 2b; 49). We analyze 30-year (climatological) averages 125 (1975-2004) using HadGEM2-ES forcing; note that the spe-126 cific forcing chosen does not appear to influence model-based 127 functional relationships (see Materials and Methods section). 128 We analyze the following model variables: precipitation P129 (the available water; equal for all models), net radiation N (a 130 proxy for the available energy), actual evapotranspiration E_a , 131 groundwater recharge R, and total runoff Q (three key water 132 fluxes), all converted to mm/y. Details on data processing are 133 described in the Materials and Methods section. 134

We address the following three research questions:

1) To what extent do key water fluxes simulated by global water models disagree with each other, and how does the disagreement vary spatially?

135

136

137

138

139

140

141

145

- 2) How do global water models translate forcing variables into key water fluxes, i.e. which functional relationships do they represent?
- 3) Can we use data and existing knowledge to derive functional relationships and use them to constrain expected model behavior?

Results

Research question 1: Global model disagreement. We first 146 assess how model outputs (actual evapotranspiration, ground-147

water recharge, total runoff) vary globally by calculating the 148 coefficient of variation (CoV) between the 8 models per grid 149 cell, shown in Figure 1a-c (see Materials and Methods for 150 151 details). Actual evapotranspiration (Figure 1a) shows low 152 CoV values suggesting reasonable agreement between models (mean grid cell CoV = 0.23), with slightly higher CoV val-153 ues in some mountainous regions (e.g. Himalaya) and cold 154 regions (e.g. northern Russia). Groundwater recharge (Figure 155 1b) shows many white spaces suggesting strong disagreement 156 (mean grid cell CoV = 1.17). The highest CoV values are 157 found in dry regions (e.g. Australia, outer-tropical Africa) 158 and in cold regions (e.g. large parts of continental Asia and 159 North America). Total runoff (Figure 1c) shows some white 160 spaces suggesting moderate disagreement (mean grid cell CoV 161 = 0.54). The highest CoV values are found in dry regions (e.g. 162 Australia, outer-tropical Africa, Central Asia). Note that the 163 CoV can be high even if the absolute differences are small, so 164 that inter-model differences might be exaggerated in very dry 165 regions. We thus show maps of the standard deviation in the 166 Supporting Information (Figures S5-7). 167

When exploring inter-model differences, it is useful to assess 168 if strong model disagreement (high CoV values) is due to high 169 ensemble uncertainty (all models disagree with each other) or 170 because one individual model deviates strongly. Figure 1d-f 171 shows which model deviates most from the ensemble mean, 172 indicating which model dominates the ensemble spread (i.e. 173 the CoV shown in Figure 1a-c). While this does not tell us 174 which models perform better or worse, it indicates whether 175 there is a single model that is consistently different (in a 176 certain region). We find that different models deviate from 177 the ensemble mean in different places. There is not one model 178 that consistently deviates the most for a specific variable (the 179 highest fraction of grid cells dominated by a single model 180 for E_a , R, and Q, is 12%, 14%, and 13%, respectively), but 181 mostly multiple models deviate similarly strongly from the 182 ensemble mean (for E_a , R, and Q, it is 31%, 28%, and 34%, 183 respectively). Overall, there is little agreement between the 184 maps and we cannot single out one model that consistently 185 deviates the most for all fluxes over a large region (Figures 186 1d-f show different patterns). 187

Research questions 2 and 3: Functional relationships. We 188 can visually assess relationships between forcing (P, N) and 189 response variables (E_a, R, Q) by inspecting scatter plots, ex-190 emplarily shown for some variable combinations and some 191 models in Figure 2a-c. To facilitate the comparison of multiple 192 models, we use Spearman rank correlations ρ_s as a summary 193 metric. A high rank correlation (close to 1) indicates that 194 spatial variability in the forcing variable (e.g. precipitation) 195 almost completely explains spatial variability in the output 196 variable (e.g. groundwater recharge). This corresponds to a 197 198 scatter plot with a tight relationship, as in Figure 2b for H08. A low rank correlation indicates that other factors also matter 199 (e.g. model parameters, other input data). This corresponds 200 to a scatter plot with a scattered relationship, as in Figure 201 2b for PCR-GLOBWB. It is important to note that a high 202 correlation is not a measure of goodness of fit. The correlations 203 simply characterize the strength of the relationship between 204 205 forcing and response variables.

Besides comparing functional relationships between models,
 we compare the models to functional relationships based on
 several observational datasets and the semi-empirical equation

3

introduced by Budyko (50), listed in Table 1.

209

Having found clear differences between major climate zones 210 (Figure 1), we divide the world into four climatic regions (see 211 Materials and Methods for details). Based on the aridity index 212 (here defined as N/P), a place is categorized as either wet 213 (N/P < 1) or dry (N/P > 1). Based on how many days 214 per year fall below a 1°C temperature threshold, a place is 215 categorized as either cold (more than one month below 1° C) 216 or warm (less than one month below 1°C). This results in four 217 categories: wet-warm (18% of modeled area), wet-cold (15%), 218 dry-cold (24%), and dry-warm (43%), shown in Figure 2d. 219

Precipitation and evapotranspiration There are some differences 220 in magnitude, but the ranking between the climate regions is 221 the same for all models (Figure 3a and Table S3 in the Support-222 ing Information). We find the strongest P- E_a -relationships in 223 dry-warm places (ρ_s ranges from 0.90-0.98) and the weakest 224 P- E_a -relationships in wet-warm places ($\rho_s: 0.57$ -0.73). The 225 Budyko equation (50) predicts higher correlations overall, but 226 the same ranking. This reflects the fact that in dry (i.e. water-227 limited) places, precipitation primarily evaporates, while in 228 wet (i.e. energy-limited) places, evapotranspiration is lim-229 ited by the available energy. Contrary to the models and the 230 Budyko equation, FLUXCOM data (51) show higher values 231 for wet-cold than for dry-cold places (Figure 3a and Table 1). 232

Net radiation and evapotranspiration Most of the models show 233 similar magnitudes and a similar ranking between the climate 234 regions (Figure 3b and Table S3). We find the strongest $N-E_a$ -235 relationships in wet-cold places (ρ_s : 0.56-0.96) and the weakest 236 *N*-*E*_a-relationships in dry-cold places (ρ_s : 0.26-0.70). This 237 agrees with the Budyko equation and reflects the fact that in 238 wet (i.e. energy-limited) places, net radiation is the primary 239 control on actual evapotranspiration, while in dry (i.e. water-240 limited) places, net radiation is a less strong control. There 241 are, however, differences in the ranking for some of the models, 242 especially for dry-warm places. For CWatM and WaterGAP2 243 dry-warm and not dry-cold places show the lowest correlation 244 (0.12 and 0.41, respectively), and for MATSIRO dry-warm 245 places show the highest correlation (0.70). FLUXCOM data 246 show very high correlations for all regions (0.79-0.94; higher 247 than most models), even for dry-cold regions (0.79), which 248 show the lowest correlations in the models (0.26-0.70) and for 249 Budyko (0.59) (Figure 3b and Table 1). 250

Precipitation and groundwater recharge The models show little 251 agreement in their P-R-relationships (Figure 3c and Table 252 S3). The largest variability is seen in dry-cold places (ρ_s : 0.35-253 (0.84), followed closely by dry-warm (0.48-0.95) and wet-cold 254 (0.47-0.90) places, while the lowest variability is seen in wet-255 warm places (0.70-0.88). CLM4.5, MATSIRO, WaterGAP2 256 and H08 show the highest correlations overall (0.71-0.95 across 257 all climate regions), while other models (CWatM, JULES-W1, 258 LPJmL, PCR-GLOBWB) show lower and more variable cor-259 relations (0.35-0.85). Some models show a clear difference 260 between the climate regions (e.g. JULES-W1; ρ_s : 0.35-0.85), 261 while others show little variability (e.g. MATSIRO; ρ_s : 0.76-262 (0.88). Groundwater recharge observations for Africa (52) and 263 the largest global scale groundwater recharge dataset compiled 264 up to date (53) suggest high to very high correlations (0.74-265 (0.84) in dry-warm places, which is similar to most models 266 except for PCR-GLOBWB. Observations (53) suggest no cor-267 relation (-0.04) for wet-warm places and low correlation for 268



Fig. 1. Left: maps showing the coefficient of variation, calculated per grid cell as the standard deviation divided by the mean of the 8 models. for different water fluxes: actual evapotranspiration (a), groundwater recharge (b), and total runoff (c). Lighter areas ("white spaces"; see 36) indicate high CoV values and thus show where models disagree most. Right: maps showing which model deviates most from the ensemble mean for each grid cell and thus contributes the most to the CoV shown in (a)-(c) for different water fluxes: actual evapotranspiration (d), groundwater recharge (e), and total runoff (f). Dark gray areas in (d)-(f) indicate that multiple models deviate similarly strongly from the ensemble mean. Empty, white areas in (d)-(f) indicate that no model deviates strongly from the ensemble mean. The percentages shown in (d)-(f) refer to the fraction of grid cells (and not land area) covered by each model. Greenland is masked out for the analysis.



Fig. 2. Scatter plots that exemplary show functional relationships for selected models and variables. Due to space constraints, we focus on a few examples with differing relationships. Scatter plots for all variable pairs are shown in Figures S18-23 in the Supporting Information. Variables shown are (a) precipitation and actual evapotranspiration for LPJmL and MATSIRO, (b) precipitation and groundwater recharge for H08 and PCR-GLOBWB, and (c) precipitation and total runoff for WaterGAP2 and CWatM. Each dot represents one grid cell and is based on the 30-year average of each flux. Spearman rank correlations ρ_s measure the strength of the relationship between forcing and response variables and are calculated for all grid cells within a climate region. Dots are colored according to the regions shown in (d). The definition of the climate regions can be found in the Materials and Methods section. The dashed line shows the 1:1 line, indicating the water limit assuming all water is supplied by precipitation. Some models have grid cells that exceed this water limit, for instance due to water transfers from neighboring cells. Detailed model-specific explanations are given in the Supporting Information.

dry-cold places (0.28), which disagrees with all models (Figure 3c and Table 1).

Net radiation and groundwater recharge Most models show clear 271 differences in magnitude and in ranking between the cli-272 mate regions, but there is no common pattern in their N-273 R-relationships (Figure 3d and Table S3). We find the largest 274 variability in dry-cold places (ρ_s : -0.35-0.51) and the lowest 275 276 variability in wet-warm places (0.23-0.54). The mostly positive rank correlation between groundwater recharge and net 277 radiation suggests that higher net radiation is associated with 278 more groundwater recharge. This is counter-intuitive, but can 279 be explained by the positive correlation between net radiation 280 and precipitation (ρ_s : 0.48-0.77 for the four climate regions; 281 see Figure S1 in the Supporting Information). 282

Precipitation and total runoff The models show mixed agree-283 ment in their *P*-*Q*-relationships (Figure 3e and Table S3). 284 LPJmL, H08, and CWatM show very high correlations (0.77-285 0.95) for all climate regions, while WaterGAP2 and PCR-286 287 GLOBWB show the lowest correlations overall (0.52-0.75). Almost all models show higher correlations for warm places 288 (0.71-0.95) than for cold places (0.52 to 0.82), so the primary 289 distinction here is between warm and cold, and not between 290 wet and dry as for evapotranspiration. This ranking agrees 291 with the Budyko equation and GRUN (an observation-based 292 293 global gridded runoff dataset; 54), but not with GSIM (data 294 from the global streamflow and metadata archive; 55, 56). The Budyko equation predicts high correlations for all climate 295 regions (0.87-0.99), while GSIM (0.73-0.89) and GRUN (0.28-296 0.93) show more variability and a different ranking between 297

the climate regions. Most disagreement can be found in drycold places, both between models (ρ_s : 0.52-0.82) and between GRUN (0.28), GSIM (0.89) and Budyko (0.90) (Figure 3e and Table 1).

Net radiation and total runoff The ranking is mostly consistent 302 and all models show rather large variability in their N-Q-303 relationships between the climate regions (Figure 3f and Table 304 S3). Dry-cold places have low to negative correlations (-0.18-305 (0.14), wet-warm and wet-cold places have low to medium 306 correlations (0.16-0.50), and dry-warm places have the highest 307 correlations overall (0.12-0.73). This tendency agrees with 308 the Budyko equation (Figure 3f and Table 1). Similar to 309 groundwater recharge, the mostly positive correlation between 310 total runoff and net radiation suggests that higher net radiation 311 is associated with more runoff, which can be explained by the 312 positive correlation between net radiation and precipitation. 313

Discussion

To what extent do key water fluxes simulated by global water 315 models disagree with each other, and how does the disagree-316 ment vary spatially?. Overall, we find the strongest model 317 disagreement for groundwater recharge (mean grid cell CoV 318 = 1.17), some disagreement for total runoff (mean grid cell 319 CoV = 0.54), and the best agreement for evapotranspiration 320 (mean grid cell CoV = 0.23). This is not unexpected, as 321 groundwater recharge is arguably the least understood process 322 (53, 58) and it is often represented in a rather simplified way 323 in the models (59). Another reason might be that – averaged 324 globally – actual evapotranspiration is the largest flux and 325

314



Fig. 3. Spearman rank correlations ρ_s between forcing variables (precipitation, net radiation) and water fluxes (actual evapotranspiration, groundwater recharge, and total runoff), divided into different climate regions. Net radiation for LPJmL and PCR-GLOBWB is not available and is estimated as the median of the other models (per grid cell). The lines connecting the dots are only there as a visual aid. The numbered triangles show observation-based rank correlations, with numbers indicating the corresponding data source (see Table 1). Observation-based rank correlations are only shown if they are based on more than 50 data points.

Table 1. Spearman rank correlations ρ_s between forcing variables and water fluxes and number of observations based on different observation-based datasets and the Budyko equation. The percentage of grid cells per climate region is given in brackets. The Budyko equation was forced per grid cell with the same forcing as the models (indicated by *), and thus covers approximately the same extent as the models (except for cells with negative net radiation). The gridded datasets (FLUXCOM, GRUN) are available at the same resolution as the models and thus also cover approximately the same extent (except for non-vegetated areas in the case of FLUXCOM). This is indicated by *m.e.* for model extent. For datasets without matching precipitation data, we used GSWP3 reanalysis data. *Number* corresponds to the numbers used in Figure 3. FLUXNET rank correlations and the MacDonald rank correlation for the wet-warm region are shown in brackets because of the very small sample sizes (50 or less); they are not shown in Figure 3. Dashes (-) indicate that correlations could not be calculated because no observations were available. Details can be found in the Materials and Methods section.

| Water flux | Forcing | Source | Number | Wet-warm (15%) | | Wet-cold (23%) | | Dry-cold (28%) | | Dry-warm (34%) | |
|------------|---------|----------------|--------|----------------|-------|----------------|-------|----------------|-------|----------------|-------|
| | | | | $ ho_s$ | Count |
| E_a | P | Budyko* (50) | 1 | 0.84 | m.e. | 0.83 | m.e. | 0.98 | m.e. | 1.00 | m.e. |
| E_a | P | FLUXCOM (51) | 2 | 0.57 | m.e. | 0.75 | m.e. | 0.67 | m.e. | 0.89 | m.e. |
| E_a | P | FLUXNET (57) | - | (0.72) | 13 | (0.36) | 50 | (0.57) | 15 | (0.51) | 47 |
| E_a | N | Budyko* (50) | 1 | 0.95 | m.e. | 0.99 | m.e. | 0.59 | m.e. | 0.79 | m.e. |
| E_a | N | FLUXCOM (51) | 2 | 0.93 | m.e. | 0.94 | m.e. | 0.79 | m.e. | 0.91 | m.e. |
| E_a | N | FLUXNET (57) | - | (0.67) | 13 | (0.42) | 50 | (0.72) | 15 | (0.41) | 47 |
| R | P | MacDonald (52) | 3 | (0.0) | 4 | - | 0 | - | 0 | 0.84 | 130 |
| R | P | Moeck (53) | 4 | -0.05 | 234 | 0.66 | 83 | 0.29 | 100 | 0.74 | 4772 |
| Q | P | Budyko* (50) | 1 | 0.94 | m.e. | 0.87 | m.e. | 0.90 | m.e. | 0.99 | m.e. |
| Q | P | GSIM (55, 56) | 5 | 0.73 | 1438 | 0.86 | 1255 | 0.89 | 593 | 0.82 | 1207 |
| Q | P | GRUN (54) | 6 | 0.86 | m.e. | 0.74 | m.e. | 0.27 | m.e. | 0.94 | m.e. |
| Q | N | Budyko* (50) | 1 | 0.45 | m.e. | 0.42 | m.e. | 0.11 | m.e. | 0.69 | m.e. |

thus its relative variability might be smaller than that of total 326 runoff and especially groundwater recharge. The models dis-327 agree particularly in dry regions (e.g. Australia, outer-tropical 328 Africa) and in cold regions (e.g. most of continental Asia and 329 North America). This echoes existing literature (e.g. 1, 26, 60) 330 and highlights the need for model refinement in dry and/or 331 cold regions, which are under-researched and strongly affected 332 by climate change (61, 62). 333

The disagreement between global water models underscores 334 the importance of using a model ensemble to account for model 335 uncertainty (6, 8, 10, 11, 24, 63). However, creating such an 336 ensemble is not straightforward (64). The use of a simple 337 ensemble mean for each grid cell leads to higher correlations 338 overall, sometimes even higher than any individual model 339 (e.g. Figure 3c for wet-warm and wet-cold regions). This 340 is because averaging smooths out variability in the forcing-341 response relationships compared to individual models (see 342 Figure S26). This finding challenges, in line with Zaherpour et 343 al. (60), the notion that the (unweighted) ensemble mean leads 344 to more robust model estimates. Assigning meaningful weights 345 to models or removing models from the ensemble (1, 65) is 346 also not straightforward. Since no model consistently deviates 347 the most, and since the most deviating models differ between 348 the three fluxes (see Figure 1d-f), model weights would have 349 to vary spatially and for each flux. Without more in-depth 350 analyses it is probably best to use the ensemble spread to 351 capture a wide range of behaviors, while acknowledging that 352 even the ensemble spread might not capture all epistemic 353 uncertainties (66). 354

How do global water models translate forcing variables into
key water fluxes, i.e. which functional relationships do they
represent?. Our evaluation shows that models differ substantially in the way they translate forcing variables into key water
fluxes (see Figure 3 and Table S3).

For precipitation and actual evapotranspiration, the models 360 show the same ranking between climate regions and rather 361 small differences in magnitude, indicating that actual evapo-362 transpiration is strongly constrained by the available water 363 for all models. We find more variability for net radiation 364 and evapotranspiration, with CWatM and WaterGAP2 show-365 ing particularly low correlations for dry-warm regions (0.12) 366 and 0.41, respectively), while all other models show much 367 higher correlations (0.69-0.87). There is no obvious corre-368 spondence between the potential evapotranspiration scheme 369 used (e.g. Priestley-Taylor for LPJmL and WaterGAP2, or 370 Penman-Monteith for JULES-W1 and CWatM) and the rank 371 correlations, implying that other factors also play an impor-372 tant role (see also 27, 63). It is worth noting that net radiation 373 differs between models (Table S4), which adds uncertainty 374 to this analysis but also highlights that the models already 375 translate the same incoming total radiation differently into 376 net radiation. This warrants a closer look in future studies, 377 since a realistic depiction of the energy balance is important 378 for climate change studies (67). 379

For precipitation and groundwater recharge, some models 380 (CLM4.5, MATSIRO, WaterGAP2 and H08) show high to very 381 high correlations (0.71-0.95) for all climate regions, suggesting 382 that precipitation is the dominant control on groundwater 383 recharge across all climate regions in these models. Other 384 models (CWatM, JULES-W1, LPJmL, PCR-GLOBWB) show 385 much lower and more variable correlations (0.35-0.85), sug-386 gesting different controls on groundwater recharge (e.g. model 387 structural decisions and parameterizations). This difference 388 can also be seen in Figure 2b, where H08 has a much tighter 389 *P-R*-relationship than PCR-GLOBWB. H08 and WaterGAP2 390 use the same approach to calculate recharge (59) and they 391 show almost the same rank correlations, indicating that the 392 functional relationships might be relatable to the model struc-393 ture in this case. We find high variability, and mostly positive 394 correlations, for net radiation and groundwater recharge. The
models probably produce more groundwater recharge in regions with higher net radiation because precipitation is also
higher in these regions. While it is difficult to interpret these
correlations, the large variability still suggests considerable
differences between models.

For precipitation and total runoff. WaterGAP2 and PCR-401 GLOBWB both show lower correlations (0.52 to 0.75) than 402 most other models (0.77 to 0.95 for CLM4.5, CWatM, H08, 403 LPJmL, and MATSIRO). This suggests clear differences in 404 how strongly total runoff is controlled by precipitation and in 405 how these models generate runoff, echoing Bierkens (68), who 406 highlighted runoff generation as main area for model improve-407 ment. This difference can also be seen in Figure 2c, where 408 CWatM shows a much tighter P-Q-relationship than Water-409 GAP2. WaterGAP2 is the only model here that is calibrated 410 against streamflow observations (59), which might explain why 411 412 it shows the lowest rank correlations for total runoff. For total runoff, it is challenging to relate model structure to functional 413 behavior, since it consists of different runoff components. Ana-414 lyzing the different runoff components individually might shed 415 more light on how different process conceptualizations (e.g. 416 dependence of surface runoff on antecedent wetness) affect 417 model behavior. Similar to groundwater recharge, we find 418 mostly positive correlations for net radiation and total runoff, 419 probably because precipitation is higher in regions with higher 420 net radiation. 421

While rank correlations expose clear differences between 422 models, our approach here should be seen as a first step in 423 quantifying functional relationships. Rank correlations only 424 measure how strongly two variables are related to each other 425 and they only capture uni-directional dependencies. Hence 426 they cannot capture all the possible differences (e.g. differences 427 in average flux magnitudes, visible in Figure 2 and quantified 428 in Table S4 in the Supplementary Information). For example, 429 since precipitation and net radiation are correlated, we should 430 not conclude that higher net radiation causes more ground-431 water recharge or runoff. More generally, rank correlations 432 themselves cannot always be easily explained by underlying 433 mechanisms. Future studies should aim at getting a better 434 mechanistic understanding of the patterns found here and 435 explore additional ways to quantify functional relationships, 436 including the use of machine learning methods. 437

Can we use data and existing knowledge to derive functional relationships and use them to constrain expected model be-

havior?. The Budyko equation (50) assumes complete depen-440 dence on aridity (here defined as N/P) and likely presents an 441 upper limit in terms of correlations. It should thus be seen 442 as a useful comparison, but not as the "correct" model, given 443 that different studies have shown that climate seasonality (69), 444 445 vegetation type (70), snow (71), and inter-catchment groundwater flow (72) can affect the long-term water balance beyond 446 aridity. Models and data reflect water- and energy-limited con-447 ditions, but tend to show lower correlations than the Budyko 448 equation (Figure 3). An exception are dry-cold regions, for 449 which FLUXCOM data show a stronger $N-E_a$ -relationship 450 and a weaker P- E_a -relationship. This might be due to a poor 451 representation of energy balance processes in cold places by 452 the Budyko equation (73) and in current models (67). 453

The Budyko framework also assumes that long-term runoff only depends on aridity. Consequently, we find higher correlations for the Budyko equation than for both datasets and most of the models. We again find large differences in drycold regions (see also 1), where GRUN shows a much weaker P-Q-relationship than the models and GSIM. 459

Recent studies have shown a strong influence of precipita-460 tion and aridity on groundwater recharge (52, 53). While our 461 results also suggest that precipitation is an important control 462 on groundwater recharge, they show that models tend to over-463 estimate the strength of that control, especially in wet-warm 464 regions and to a lesser extent in dry-cold regions. Perceptual 465 models of groundwater recharge generation usually include 466 climate, but also soil characteristics, topography, and land 467 use (74). Recently, Cuthbert et al. (75) found that local 468 hydrogeology influences *P*-*R*-relationships in Africa, though 469 this is difficult to generalize to larger regions given limitations 470 in global datasets (25). In line with those findings, our results 471 strongly suggest that models overestimate the degree to which 472 climate forcing variables control groundwater recharge. 473

We have collected several observational or observation-474 driven datasets to derive empirical forcing-response relation-475 ships, but there remain some challenges. Observation-based 476 estimates contain uncertainty, inherited from the observational 477 datasets themselves and due to small numbers of observations 478 in certain regions. Since not all datasets come with corre-479 sponding forcing and response variables, we sometimes had 480 to pair observations with other forcing datasets, which can 481 introduce additional uncertainty (see Materials and Methods 482 for an extended discussion). Looking ahead, the gaps in the 483 observation-based correlations shown in Table 1 are a first step 484 to identify regions and variables where more measurements 485 would be especially useful to constrain expected functional 486 relationships. In addition, more quality-controlled datasets 487 with uncertainty estimates (e.g 52) are critical to obtain real-488 istic uncertainty estimates for functional relationships. This 489 would ultimately allow us to obtain robust ranges of functional 490 behavior which we can benchmark our models against. 491

A global perspective for global models. Using functional re-492 lationships shifts the focus away from evaluating model per-493 formance in specific locations and from matching historical 494 records to a more diagnostic and process-oriented evaluation 495 of model behavior (76). Functional relationships allow us to 496 focus on larger-scale assessments and to explore if dominant 497 controls in the models are consistent with observations, theory 498 and expectations, i.e. our perceptual model (36). This is 499 critical for ensuring that models faithfully represent real-world 500 systems, leading to more credible projections of environmental 501 change impacts. 502

An advantage of functional relationships is that they relate 503 different locations to each other, and thus take information 504 out of its location-specific context and put it to a more large-505 scale use. However, the uneven distribution of observations 506 poses challenges if we want to derive robust relationships. For 507 example, recharge measurements have almost entirely been 508 made in warm dry regions (97% of MacDonald et al. (52) and509 92% of Moeck et al. (53)). Streamflow measurements have 510 been made more frequently in wet regions (60%) of the GSIM 511 data (55, 56) used here) and globally, there is a placement 512 bias of stream gauges towards wet regions (77), even though 513 according to our classification – short of two-thirds of the 514 global land area are dry. While there are clear reasons for 515 this spatial bias, we will have to explore how this bias affects 516 ⁵¹⁷ functional relationships and how to most effectively enlarge⁵¹⁸ our observational database.

In this study, we have focused on rank correlations between 519 long-term averages of two forcing variables and three water 520 fluxes, but this approach can easily be extended. Other vari-521 ables, including state variables or stores (e.g. soil moisture, ter-522 restrial water storage), possibly investigated at different time 523 scales (e.g. monthly), should yield additional insights. There 524 are existing metrics such as elasticities (34) that lend them-525 526 selves for such an analysis, and there is room for new methods to be developed (e.g. characterizing thresholds in forcing-527 response relationships). Expanding our range of functional 528 relationships, constrained by various observational datasets 529 and expert knowledge, might eventually give us a knowledge 530 base of realistic system behavior that can be used to evaluate 531 models and diagnose model deficiencies, comparable to the 532 use of emergent constraints in climate modeling (78). 533

534 More generally, functional relationships invite us to think about how the global water cycle functions, what we know, 535 what we do not know, and what that means for a future under 536 climate change (36). Our results here suggest that improved 537 process understanding will be particularly important for energy 538 balance processes, recharge processes, and generally in dry 539 and/or cold regions. So how can we improve our process 540 understanding? In 1986, Eagleson (79) stated that "science 541 advances on two legs, analysis and experimentation, and at 542 any moment one is ahead of the other. At the present time 543 advances in hydrology appear to be data limited". For some 544 processes, this still seems to be the case. But clearly, we have 545 a wealth of data available and might ask ourselves: are we 546 extracting enough information from the observations we have? 547 Are there hydrological regularities yet to be found (80)? Even 548 if the search for such regularities is challenging, it might be a 549 fruitful and exciting endeavor for global hydrology. 550

551 Materials and Methods

Model data retrieval and processing. We analyzed 30-year (clima-552 tological) averages (1975-2004) from 8 global water models (49): 553 CLM4.5 (41), CWatM (42), H08 (43), JULES-W1 (44), LPJmL 554 (45), MATSIRO (46), PCR-GLOBWB (47), and WaterGAP2 (48). 555 The model simulations were carried out following the ISIMIP 2b 556 protocol and here we used model outputs forced with the Earth 557 system model HadGEM2-ES under historical conditions (historical 558 climate and CO_2 concentrations). We used precipitation P (ISIMIP 559 variable name pr), net radiation N, actual evapotranspiration E_a 560 (ISIMIP variable name evap), groundwater recharge R (ISIMIP 561 variable name qr) and total runoff Q (ISIMIP variable name qtot). 562 Note that Q here refers to runoff generated on the land fractions 563 (and not surface water bodies) of each grid cell and does not include 564 upstream inflows, which allows for comparison to grid cell P. P. 565 566 E_a , R, and Q were downloaded from https://data.isimip.org/. Net radiation N is not an official ISIMIP output and was provided by 567 the individual modeling groups. It is not available for all models, so 568 we used the ensemble mean per grid cell for models without N data. 569 We converted all fluxes to mm/y and removed E_a values larger than 570 10000 mm/y and set R values smaller than 0 to 0. A more detailed 571 description is given in the Supporting Information. 572

CoV and most deviating model maps. For each grid cell, we calculated the coefficient of variation (CoV) by dividing the standard deviation by the mean using the 8 model outputs. Maps of the standard deviation are shown in the Supporting Information (Figures S5-7). To see which model dominates the ensemble spread, we checked for each grid cell which model shows the largest absolute difference (denoted by d_1) from the ensemble mean (denoted by μ).

9

To see if multiple models dominate the ensemble spread, we also 580 checked for each grid cell which model shows the second largest 581 absolute difference (denoted by d_2) from the ensemble mean. If 582 the relative difference between the largest and the second largest 583 difference is less than 20%, i.e. $(d_1 - d_2)/d_1 < 0.2$, the grid cell falls into the category "multiple". If the relative difference between 584 585 the most deviating model and the ensemble mean is less than 20%, 586 i.e. $d_1/\mu < 0.2$, the grid cell is counted as having no most deviating 587 model (empty areas on Figure 1d-f). 588

Functional relationships. As a metric for the strength of the func-589 tional relationships between model inputs and outputs, we use 590 Spearman rank correlations ρ_s for each climate region. The Spear-591 man rank correlation is a measure of the monotonicity between two 592 variables and it is robust to outliers. We use the following categories 593 for correlations: negative (<0), no to low correlation (0 to 0.25), 594 medium correlation (0.25-0.5), high correlation (0.5-0.75), very high 595 correlation (0.75-1.0). 596

Climate regions Based on the aridity index (here defined as N/P), 597 a place is categorized as either wet (N/P < 1) or dry (N/P > 1). 598 Note that we used the ensemble median for N. Based on how 599 many days per year fall below a 1°C temperature threshold, a place 600 is categorized as either cold (more than one month below 1°C) 601 or warm (less than one month below 1°C). This results in four 602 categories: wet-warm (15% of model grid cells / 18% of modeled 603 area), wet-cold (23% / 15%), dry-cold (28% / 24%), and dry-warm 604 (34% / 43%). To test how different decisions affect our climate 605 region classification, we also used the ensemble median of potential 606 evapotranspiration E_p (partially downloaded, partially provided by 607 the modeling groups) to calculate the aridity index (E_p/P) , and 608 we used a different threshold for our warm/cold distinction. This 609 resulted in little differences overall, as can be seen in the Supporting 610 Information (Figure S16). 611

Observational datasets and theory. For E_a , we used FLUXCOM 612 data (51) (RS monthly 0.5° from 2001-2015) paired with GSWP3 P 613 data (81) (downloaded from https://data.isimip.org/), and FLUXNET 614 data (57) which include matching P data. For R, we used data 615 from MacDonald et al. (52) which include matching P data, and 616 data from Moeck et al. (53) paired with GSWP3 P data (81). For 617 Q, we used GRUN data (54) from 1985-2004 paired with GSWP3 P618 data (81), and GSIM data (55, 56) from catchments with areas from 619 $250-25000 \text{ km}^2$ with minimum 10y of data to ensure a sufficient 620 number of catchments that do not differ too much in size from the 621 model grid cells. We paired GSIM data with catchment-averaged 622 MSWEP P data (82), which were calculated by Stein et al. (83). 623

To obtain theory-based estimates for E_a and Q, we forced the Budyko (50) equation (Eq.1) with HadGEM2-ES P and ensemble median N from the ISIMIP 2b models analyzed here.

$$\frac{E_a}{P} = \sqrt{\frac{N}{P}} \tanh\left(\frac{P}{N}\right) \left(1 - \exp\left(-\frac{N}{P}\right)\right)$$
[1]

More details on data processing and quality checks can be found in the Supporting Information.

Extended discussion on model forcing and scenario uncertainty. ${\rm The}$ choice of forcing product and differences in the treatment of human 627 influences (e.g. water use and dams) might affect the functional 628 relationships exhibited by the models. To get an idea how much un-629 certainty this introduces, we calculated correlations using WATCH-630 WFDEI forcing with either variable historical conditions (varsoc) or 631 no human influences (nosoc) for WaterGAP2 and PCR-GLOBWB, 632 carried out following the ISIMIP 2a protocol. The results, shown in 633 the Supporting Information, stay essentially the same, suggesting 634 that the model-based correlations are robust signatures of model 635 behavior. 636

Extended discussion on data uncertainty. Since not all datasets come with matching P data, we sometimes paired the observations with GSWP3 reanalysis data (81). To get an idea how much uncertainty this introduces, we calculated rank correlations using different Pdata sources. Correlations calculated using the MacDonald et al. (52) R data with either GSWP3 P data or the accompanying P data are very similar for dry-warm places (0.83 and 0.84; see Supporting

625 626

624

- Information). Using HadGEM2-ES P (the model forcing) data 644
- instead of GSWP3 P data to calculate correlations with FLUXCOM 645
- E_a (51), Moeck R (53), and GRUN Q (54), respectively, results 646

647 in virtually no differences (results are shown in the Supporting

- information). This indicates that the correlations are robust, likely 648
- because rank correlations remain stable as long as relative differences 649
- between forcing values per grid cell stay the same. 650
- Code and data availability. Model outputs can be accessed via the 651 ISIMIP website (https://www.isimip.org/). Observational datasets can 652 be accessed via the references shown in Table 1. Multi-annual 653
- averages and rank correlations will be uploaded to a repository 654
- and code can be found at https://github.com/HydroSysPotsdam/Global_ 655
- model_evaluation (DOI for both will be created upon acceptance). 656

657 **ACKNOWLEDGMENTS.** SJG, RR, LS, and TW acknowledge support from the Alexander von Humboldt Foundation in the frame-658 659 work of the Alexander von Humboldt Professorship endowed by the German Federal Ministry of Education and Research (BMBF). 660 This publication is based upon work from COST Action CA19139 -661 662 PROCLIAS, supported by COST (European Cooperation in Science and Technology, https://www.cost.eu). We thank Anneke Tombrägel 663 664 for help with processing the FLUXNET data and Petra Döll for providing helpful comments on the manuscript. We also thank the 665 666 ISIMIP team for their continued efforts within the ISIMIP project.

- A Gädeke, et al., Performance evaluation of global hydrological models in six large Pan-Arctic 667 1. watersheds, Clim, Chang, 163, 1329-1351 (2020). 668
- D Gerten, et al., Asynchronous exposure to global warming: freshwater resources and ter-669 2. restrial ecosystems. Environ. Res. Lett. 8, 034032 (2013) Publisher: IOP Publishing 670
- 671 3. NW Arnell, SN Gosling, The impacts of climate change on river flood risk at the global scale 672 Clim. Chang. 134, 387-401 (2016).
- P Greve, et al., Global assessment of water challenges under uncertainty in water scarcity 673 4 674 proiections. Nat. Sustain. 1, 486-494 (2018) Number: 9 Publisher: Nature Publishing Group
- 5. 675 J Schewe, et al., State-of-the-art global models underestimate impacts from climate extremes 676 Nat. Commun. 10, 1005 (2019) Number: 1 Publisher: Nature Publishing Group.
- 677 W Thiery, et al., Intergenerational inequities in exposure to climate extremes. Science 374. 6. 678 158–160 (2021) Publisher: American Association for the Advancement of Science
- 679 7. IPCC, Climate Change 2022: Impacts, Adaptation, and Vulnerability. Contribution of Working 680 Group II to the Sixth Assessment Report of the Intergovernmental Panel on Climate Change 681 [H.-O. Pörtner, D.C. Roberts, M. Tignor, E.S. Poloczanska, K. Mintenbeck, A. Alegría, M. 682 Craig, S. Langsdorf, S. Löschke, V. Möller, A. Okem, B. Rama (eds.)]. (Cambridge University Press), (2022). 683
 - J Schewe, et al., Multimodel assessment of water scarcity under climate change. Proc. Natl. 8. Acad. Sci. 111, 3245-3250 (2014) Publisher: Proceedings of the National Academy of Sciences
- L Gudmundsson, et al., Globally observed trends in mean and extreme river flow attributed 9 to climate change. Science 371, 1159-1162 (2021) Publisher: American Association for the Advancement of Science. 689
 - 10. Y Pokhrel, et al., Global terrestrial water storage and drought severity under climate change Nat. Clim. Chang. 11, 226–233 (2021) Number: 3 Publisher: Nature Publishing Group.
 - 11. R Reinecke, et al., Uncertainty of simulated groundwater recharge at different global warming evels: a global-scale multi-model ensemble study. Hydrol. Earth Syst. Sci. 25, 787-810 (2021) Publisher: Copernicus GmbH.
 - IGRAC, Global Groundwater Information System (2022) Published: ggis.un-igrac.org. 12.
 - RW Hofste, et al., Aqueduct 3.0: Updated decision-relevant global water risk indicators. World 13. Resour. Inst. pp. 1-53 (2019) Publisher: World Resources Institute.
- 14. J Sheffield, et al., A Drought Monitoring and Forecasting System for Sub-Sahara African Water Resources and Food Security. Bull. Am. Meteorol. Soc. 95, 861-882 (2014) Publisher: 699 700 American Meteorological Society Section: Bulletin of the American Meteorological Society.
 - P Döll, K Fiedler, Global-scale modeling of groundwater recharge. Hydrol. Earth Syst. Sci. 15. 12, 863-885 (2008) Publisher: Copernicus GmbH.
- Y Wada, et al., Global depletion of groundwater resources. Geophys. Res. Lett. 37 (2010) 703 16. _eprint: https://onlinelibrary.wiley.com/doi/pdf/10.1029/2010GL044571. 704
- Richey, et al., Quantifying renewable groundwater 705 stress (2015) GRACE. Water Resour. Res. 51. 5217-5238 _eprint: 706 https://onlinelibrary.wiley.com/doi/pdf/10.1002/2015WR017349. 707
- IEM de Graaf, et al., A global-scale two-layer transient groundwater model: Development and 708 18. application to groundwater depletion. Adv. Water Resour. 102, 53-67 (2017). 709
- R Reinecke, et al., Challenges in developing a global gradient-based groundwater model 19. 710 (G³M v1.0) for the integration into a global hydrological model. *Geosci, Model, Dev.* 12. 711 2401-2418 (2019) Publisher: Copernicus GmbH. 712
- 713 20. R Dankers, et al., First look at changes in flood hazard in the Inter-Sectoral Impact Model Intercomparison Project ensemble. Proc. Natl. Acad. Sci. 111, 3257-3261 (2014) Publisher: 714 715 Proceedings of the National Academy of Sciences.
- L Samaniego, et al., Anthropogenic warming exacerbates European soil moisture droughts 21 716 Nat. Clim. Chang. 8, 421-426 (2018) Number: 5 Publisher: Nature Publishing Group. 717
- 22. C Prudhomme, et al., Hydrological droughts in the 21st century, hotspots and uncertainties 718 from a global multimodel ensemble experiment, Proc. Natl. Acad. Sci. 111, 3262-3267 (2014) 719 720 Publisher: Proceedings of the National Academy of Sciences.

- 23. HE Beck, et al., Global evaluation of runoff from 10 state-of-the-art hydrological models. Hy-721 drol. Earth Syst. Sci. 21, 2881-2903 (2017) Publisher: Copernicus GmbH. 722
- 24. BR Scanlon, et al., Global models underestimate large decadal declining and rising water 723 storage trends relative to GRACE satellite data. Proc. Natl. Acad. Sci. 115, E1080-E1089 724 (2018) Publisher: Proceedings of the National Academy of Sciences. 725

726

727

728

729

730

731

732

733

734

735

736

737

738

739

740

741

742

743

744

745

746

747

748

749

750

751

752

753

754

755

756

757

758

759

760

761

762

763

764

765

766

767

768

769

770

771

772

773

774

775

776

777

778

779

780

781

782

783

784

785

786

787

788

789

790

791

792

793

794

795

796

797

798

799

800

801

802

803

804

- C West, et al., Ground truthing global-scale model estimates of groundwater recharge across 25. Africa. Sci. The Total. Environ. 858, 159765 (2023).
- I Giuntoli, JP Vidal, C Prudhomme, DM Hannah, Future hydrological extremes: the uncer-26 tainty from multiple global climate and global hydrological models. Earth Syst. Dyn. 6, 267-285 (2015) Publisher: Copernicus GmbH.
- R Wartenburger, et al., Evapotranspiration simulations in ISIMIP2a-Evaluation of spatio-27. temporal characteristics with a comprehensive ensemble of independent datasets. Environ. Res. Lett. 13, 075001 (2018) Publisher: IOP Publishing.
- 28. A Hartmann, T Gleeson, Y Wada, T Wagener, Enhanced groundwater recharge rates and altered recharge sensitivity to climate variability through subsurface heterogeneity. Proc. Natl. Acad. Sci. 114, 2842-2847 (2017) Publisher: Proceedings of the National Academy of Sciences.
- 29. T Gleeson, et al., GMD perspective: The quest to improve the evaluation of groundwate representation in continental- to global-scale models. Geosci. Model. Dev. 14, 7545-7571 (2021) Publisher: Copernicus GmbH.
- 30. Wagener, R Reinecke, F Pianosi. On the evaluation climate impact models. WIREs Clim. Chang. 13, e772 (2022) _eprint: change https://onlinelibrary.wiley.com/doi/pdf/10.1002/wcc.772.
- RD Koster, P Milly, The Interplay between Transpiration and Runoff Formulations in Land 31. Surface Schemes Used with Atmospheric Models. J. Clim. 10 (1997).
- 32. R Knutti, MAA Rugenstein, GC Hegerl, Beyond equilibrium climate sensitivity. Nat. Geosci. 10, 727-736 (2017) Number: 10 Publisher: Nature Publishing Group.
- J NĚMEC, J SCHAAKE, Sensitivity of water resource systems to climate varia-33. Hydrol. Sci. J. 27, 327-343 (1982) Publisher: Taylor & Francis _eprint: tion. https://doi.org/10.1080/02626668209491113.
- 34. A Sankarasubramanian, RM Vogel, JF Limbrunner, Climate elasticity of stream-Water Resour. Res. 37, 1771-1781 (2001) _eprint: flow in the United States. https://onlinelibrary.wiley.com/doi/pdf/10.1029/2000WR900330.
- 35. B Niissen, GM O'Donnell, AF Hamlet, DP Lettenmaier, Hydrologic Sensitivity of Global Rivers to Climate Change, Clim, Chang, 50, 143-175 (2001).
- 36. T Wagener, et al., On doing hydrology with dragons: Realizing the value of percep-WIREs Water 8, e1550 (2021) _eprint: tual models and knowledge accumulation. https://onlinelibrary.wiley.com/doi/pdf/10.1002/wat2.1550.
- 37. RM Vogel. Sankarasubramanian, Validation watershed Α ∩f а Resour. calibration. model without Water Res. 39 (2003) _eprint: https://onlinelibrary.wiley.com/doi/pdf/10.1029/2002WB001940.
- E Kapangaziwiri, D Hughes, T Wagener, Incorporating uncertainty in hydrological predictions 38 for gauged and ungauged basins in southern Africa, Hvdrol, Sci. J. 57, 1000-1019 (2012) Publisher: Taylor & Francis eprint: https://doi.org/10.1080/02626667.2012.690881.
- 39. TJ Troy, EF Wood, J Sheffield, An efficient calibration method for continental-_eprint: scale land surface modeling. Water Resour. Res. 44 (2008) https://onlinelibrary.wiley.com/doi/pdf/10.1029/2007WR006513.
- P Greve, P Burek, Y Wada, Using the Budyko Framework for Calibrating a 40 Global Hydrological Model. Water Resour. Res. 56, e2019WR026280 (2020) _eprint: https://onlinelibrary.wiley.com/doi/pdf/10.1029/2019WR026280
- Present-day 41 w Thiery, et al., irrigation mitidates heat extremes. Res. J. Geophys. Atmospheres 122. 1403-1422 (2017) _eprint: https://onlinelibrary.wiley.com/doi/pdf/10.1002/2016JD025740.
- 42. P Burek, et al., Development of the Community Water Model (CWatM v1.04) a highresolution hydrological model for global and regional assessment of integrated water resources management. Geosci. Model. Dev. 13, 3267-3298 (2020) Publisher: Copernicus GmhH
- 43. N Hanasaki, S Yoshikawa, Y Pokhrel, S Kanae, A global hydrological simulation to specify the sources of water used by humans. Hydrol. Earth Syst. Sci. 22, 789-817 (2018) Publisher: Copernicus GmbH.
- 44 MJ Best, et al., The Joint UK Land Environment Simulator (JULES), model description - Part 1: Energy and water fluxes. Geosci. Model. Dev. 4, 677-699 (2011) Publisher: Copernicus GmbH
- J Jägermeyr, et al., Water savings potentials of irrigation systems: global simulation of pro-45. cesses and linkages. Hydrol. Earth Syst. Sci. 19, 3073-3091 (2015) Publisher: Copernicus GmbH
- 46. K Takata, S Emori, T Watanabe, Development of the minimal advanced treatments of surface interaction and runoff. Glob. Planet. Chang. 38, 209-222 (2003).
- EH Sutanudjaja, et al., PCR-GLOBWB 2: a 5 arcmin global hydrological and water resources model, Geosci, Model, Dev. 11, 2429-2453 (2018) Publisher: Copernicus GmbH.
- H Müller Schmied, et al., Variations of global and continental water balance components as 48. impacted by climate forcing uncertainty and human water use. Hydrol. Earth Syst. Sci. 20, 2877-2898 (2016) Publisher: Copernicus GmbH.
- 49. K Frieler, et al., Assessing the impacts of 1.5 °C global warming simulation protocol of the Inter-Sectoral Impact Model Intercomparison Project (ISIMIP2b). Geosci. Model. Dev. 10, 4321-4345 (2017) Publisher: Copernicus GmbH.
- MI Budyko, Climate and life, (Academic Press), Technical Report 18 (1974).
- 51. M Jung, et al., The FLUXCOM ensemble of global land-atmosphere energy fluxes. Sci. Data 6, 74 (2019) Number: 1 Publisher: Nature Publishing Group.
- AM MacDonald, et al., Mapping groundwater recharge in Africa from ground observations 52. and implications for water security. Environ. Res. Lett. 16, 034012 (2021) Publisher: IOP Publishing.
- 53. C Moeck, et al., A global-scale dataset of direct natural groundwater recharge rates; A review of variables, processes and relationships. Sci. The Total. Environ. 717, 137042 (2020).

684

685

686

687

688

690

691

692

693

694

695

696

697

698

701

702

non peer-reviewed preprint submitted to PNAS

- G Ghiggi, V Humphrey, SI Seneviratne, L Gudmundsson, GRUN: an observation-based
 global gridded runoff dataset from 1902 to 2014. *Earth Syst. Sci. Data* 11, 1655–1674 (2019)
 Publisher: Copernicus GmbH.
- HX Do, L Gudmundsson, M Leonard, S Westra, The Global Streamflow Indices and Metadata
 Archive (GSIM) Part 1: The production of a daily streamflow archive and metadata. *Earth Syst. Sci. Data* 10, 765–785 (2018) Publisher: Copernicus GmbH.
- L Gudmundsson, HX Do, M Leonard, S Westra, The Global Streamflow Indices and Metadata
 Archive (GSIM) Part 2: Quality control, time-series indices and homogeneity assessment.
 Earth Syst. Sci. Data 10, 787–804 (2018) Publisher: Copernicus GmbH.
- 67. G Pastorello, et al., The FLUXNET2015 dataset and the ONEFlux processing pipeline for
 eddy covariance data. *Sci. Data* 7, 225 (2020) Number: 1 Publisher: Nature Publishing
 Group.
- TR Green, et al., Beneath the surface of global change: Impacts of climate change on groundwater. J. Hydrol. 405, 532–560 (2011).
- S9. CE Telteu, et al., Understanding each other's models: an introduction and a standard representation of 16 global water models to support intercomparison, improvement, and communication. *Geosci. Model. Dev.* 14, 3843–3878 (2021) Publisher: Copernicus GmbH.
- J Zaherpour, et al., Worldwide evaluation of mean and extreme runoff from six global-scale
 hydrological models that account for human impacts. *Environ. Res. Lett.* 13, 065015 (2018)
 Publisher: IOP Publishing.
- F Felfelani, Y Wada, L Longuevergne, YN Pokhrel, Natural and human-induced terrestrial water storage change: A global analysis using hydrological models and GRACE. J. Hydrol.
 553, 105–118 (2017).
- D Zoccatelli, et al., Contrasting rainfall-runoff characteristics of floods in desert and Mediterranean basins. *Hydrol. Earth Syst. Sci.* 23, 2665–2678 (2019) Publisher: Copernicus GmbH.
- I Haddeland, et al., Multimodel Estimate of the Global Terrestrial Water Balance: Setup and
 First Results. J. Hydrometeorol. 12, 869–884 (2011) Publisher: American Meteorological
 Society Section: Journal of Hydrometeorology.
- 44. J Rougier, Ensemble Averaging and Mean Squared Error. J. Clim. 29, 8865–8870 (2016)
 Publisher: American Meteorological Society Section: Journal of Climate.
- NP Gillett, Weighting climate model projections using observational constraints. *Philos. Trans*actions Royal Soc. A: Math. Phys. Eng. Sci. 373, 20140425 (2015) Publisher: Royal Society.
- KJ Beven, et al., Epistemic uncertainties and natural hazard risk assessment Part 2: What
 should constitute good practice? *Nat. Hazards Earth Syst. Sci.* 18, 2769–2783 (2018) Publisher: Copernicus GmbH.
- PCD Milly, KA Dunne, Colorado River flow dwindles as warming-driven loss of reflective snow
 energizes evaporation. *Science* 367, 1252–1255 (2020) Publisher: American Association for
 the Advancement of Science.
- MFP Bierkens, Global hydrology 2015: State, trends, and directions. *Water Resour. Res.* 51, 4923–4947 (2015) _eprint: https://onlinelibrary.wiley.com/doi/pdf/10.1002/2015WR017173.
- 845 69. PCD Milly, Climate, soil water storage, and the average annual wa 846 ter balance. Water Resour. Res. 30, 2143–2156 (1994) _eprint:
 847 https://onlinelibrary.wiley.com/doi/pdf/10.1029/94WR00586.
- L Zhang, WR Dawes, GR Walker, Response of mean annual evapotranspiration to vegetation changes at catchment scale. *Water Resour. Res.* 37, 701–708 (2001) _eprint: https://onlinelibrary.wiley.com/doi/pdf/10.1029/2000WR900325.
- WR Berghuijs, RA Woods, M Hrachowitz, A precipitation shift from snow towards rain leads
 to a decrease in streamflow. *Nat. Clim. Chang.* 4, 583–586 (2014) Number: 7 Publisher:
 Nature Publishing Group.
- Y Liu, T Wagener, HE Beck, A Hartmann, What is the hydrologically effective area of a catchment? *Environ. Res. Lett.* **15**, 104024 (2020) Publisher: IOP Publishing.
- AA Meira Neto, GY Niu, T Roy, S Tyler, PA Troch, Interactions between snow cover and evaporation lead to higher sensitivity of streamflow to temperature. *Commun. Earth & Environ.* 1, 1–7 (2020) Number: 1 Publisher: Nature Publishing Group.
- BR Scanlon, RW Healy, PG Cook, Choosing appropriate techniques for quantifying groundwater recharge. *Hydrogeol. J.* p. 22 (2002).
- 75. MO Cuthbert, et al., Observed controls on resilience of groundwater to climate variability in sub-Saharan Africa. *Nature* 572, 230–234 (2019) Number: 7768 Publisher: Nature Publishing Group.
- HV Gupta, T Wagener, Y Liu, Reconciling theory with observations: elements of a diagnostic approach to model evaluation. *Hydrol. Process.* 22, 3802–3813 (2008) _eprint: https://onlinelibrary.wiley.com/doi/pdf/10.1002/hyp.6989.
- 867 77. CA Krabbenhoft, et al., Assessing placement bias of the global river gauge network. Nat.
 868 Sustain. pp. 1–7 (2022) Publisher: Nature Publishing Group.
- 78. V Eyring, et al., Taking climate model evaluation to the next level. Nat. Clim. Chang. 9, 102–
 110 (2019) Number: 2 Publisher: Nature Publishing Group.
- 79. PS Eagleson, The emergence of global-scale hydrology. Water Resour. Res. 22, 6S–14S
 (1986) _eprint: https://onlinelibrary.wiley.com/doi/pdf/10.1029/WR022i09Sp0006S.
- JCI Dooge, Looking for hydrologic laws. *Water Resour. Res.* 22, 46S–58S (1986) _eprint: https://onlinelibrary.wiley.com/doi/pdf/10.1029/WR022i09Sp0046S.
- PA Dirmeyer, et al., GSWP-2: Multimodel Analysis and Implications for Our Perception of
 the Land Surface. Bull. Am. Meteorol. Soc. 87, 1381–1398 (2006) Publisher: American
 Meteorological Society Section: Bulletin of the American Meteorological Society.
- HE Beck, et al., MSWEP V2 Global 3-Hourly 0.1° Precipitation: Methodology and Quantitative Assessment. *Bull. Am. Meteorol. Soc.* 100, 473–500 (2019) Publisher: American Meteorological Society Section: Bulletin of the American Meteorological Society.
- St. L. Stein, F. Pianosi, R. Woods, Event-based classification for global study of river flood generating processes. *Hydrol. Process.* 34, 1514–1529 (2020) _eprint: https://onlinelibrary.wiley.com/doi/pdf/10.1002/hyp.13678.